Author’s response to reviews

Title: Caries Prevalence and Dental Health of 8-12 Year-Old Children in Damascus City in Syria during the Syrian Crisis; a Cross-Sectional Epidemiological Oral Health Survey

Authors:
Muhammed Al-Huda BALLOUK (dr.muhammed.alhuda@gmail.com)
Mayssoon DASHASH (mdashash@yahoo.com)

Version: 2 Date: 11 Dec 2018

Author’s response to reviews:

Dear Professor Dr. Marlise Klein;

Have a nice day professor …

This is concerning: OHEA-D-18-00438, with the title of “Caries Prevalence and Dental Health of 8-12 Year-Old Children in Damascus City in Syria during the Syrian Crisis; a Cross-Sectional Epidemiological Oral Health Survey”. By: Muhammed Al-Huda BALLOUK, & Mayssoon DASHASH.

Here is my response to the issues raised by the reviewers:

Lynn Al-Zwaylif (Reviewer 3)

Thanks a lot for your highly valued decision and precious notes.

I added a brief note about the limitations at the end of the Discussion section as to your kind advice. It is only that the BMC Oral Health guidelines state that the Limitations section should be at the end of the text. Still, surely I added this brief limitations “hint” to the Discussion as to
your kind advice but if you think that there are any other changes that I should do then please let me know and I will be more than pleased to do them.

Again thank you very much indeed.

Very obliged to you.

As for the 4th Reviewer (Peter Bottenberg), we decline to revise our manuscript as to any of his comments.

So much thankful to you indeed for all your time and efforts.

Eager to hearing from you real real soon.

Best Regards

Yours Sincerely

Dr. Muhammed Al-Huda BALLOUK